Author's response to reviews

Title: A stimulus to define informatics and health information technology

Authors:

    William R Hersh (hersh@ohsu.edu)

Version: 7 Date: 12 May 2009

Author's response to reviews:

Fixing some final typos and reducing the length of the first run-on sentence in the abstract.